import { TypeList } from "docs-ui"

# translation

## Interfaces

- [BaseFilterable](../../translation/interfaces/translation.BaseFilterable/page.mdx)
- [Context](../../translation/interfaces/translation.Context/page.mdx)
- [CreateLocaleDTO](../../translation/interfaces/translation.CreateLocaleDTO/page.mdx)
- [CreateTranslationDTO](../../translation/interfaces/translation.CreateTranslationDTO/page.mdx)
- [EntityTypeStatistics](../../translation/interfaces/translation.EntityTypeStatistics/page.mdx)
- [FilterableLocaleProps](../../translation/interfaces/translation.FilterableLocaleProps/page.mdx)
- [FilterableTranslationProps](../../translation/interfaces/translation.FilterableTranslationProps/page.mdx)
- [FindConfig](../../translation/interfaces/translation.FindConfig/page.mdx)
- [IMessageAggregator](../../translation/interfaces/translation.IMessageAggregator/page.mdx)
- [IModuleService](../../translation/interfaces/translation.IModuleService/page.mdx)
- [ITranslationModuleService](../../translation/interfaces/translation.ITranslationModuleService/page.mdx)
- [JoinerServiceConfig](../../translation/interfaces/translation.JoinerServiceConfig/page.mdx)
- [JoinerServiceConfigAlias](../../translation/interfaces/translation.JoinerServiceConfigAlias/page.mdx)
- [LocaleDTO](../../translation/interfaces/translation.LocaleDTO/page.mdx)
- [LocaleStatistics](../../translation/interfaces/translation.LocaleStatistics/page.mdx)
- [MessageAggregatorFormat](../../translation/interfaces/translation.MessageAggregatorFormat/page.mdx)
- [RestoreReturn](../../translation/interfaces/translation.RestoreReturn/page.mdx)
- [SoftDeleteReturn](../../translation/interfaces/translation.SoftDeleteReturn/page.mdx)
- [TranslationDTO](../../translation/interfaces/translation.TranslationDTO/page.mdx)
- [TranslationStatisticsInput](../../translation/interfaces/translation.TranslationStatisticsInput/page.mdx)
- [UpdateLocaleDataDTO](../../translation/interfaces/translation.UpdateLocaleDataDTO/page.mdx)
- [UpdateLocaleDTO](../../translation/interfaces/translation.UpdateLocaleDTO/page.mdx)
- [UpdateTranslationDataDTO](../../translation/interfaces/translation.UpdateTranslationDataDTO/page.mdx)
- [UpdateTranslationDTO](../../translation/interfaces/translation.UpdateTranslationDTO/page.mdx)

## Type Aliases

- [Constructor](../../translation/types/translation.Constructor/page.mdx)
- [Event](../../translation/types/translation.Event/page.mdx)
- [EventMetadata](../../translation/types/translation.EventMetadata/page.mdx)
- [ExpandScalar](../../translation/types/translation.ExpandScalar/page.mdx)
- [FilterQuery](../../translation/types/translation.FilterQuery/page.mdx)
- [FilterQueryProperties](../../translation/types/translation.FilterQueryProperties/page.mdx)
- [FilterValue](../../translation/types/translation.FilterValue/page.mdx)
- [FilterValue2](../../translation/types/translation.FilterValue2/page.mdx)
- [FindConfigOrder](../../translation/types/translation.FindConfigOrder/page.mdx)
- [JoinerRelationship](../../translation/types/translation.JoinerRelationship/page.mdx)
- [LinkModulesExtraFields](../../translation/types/translation.LinkModulesExtraFields/page.mdx)
- [Message](../../translation/types/translation.Message/page.mdx)
- [ModuleJoinerConfig](../../translation/types/translation.ModuleJoinerConfig/page.mdx)
- [ModuleJoinerRelationship](../../translation/types/translation.ModuleJoinerRelationship/page.mdx)
- [OperatorMap](../../translation/types/translation.OperatorMap/page.mdx)
- [PrevLimit](../../translation/types/translation.PrevLimit/page.mdx)
- [Primary](../../translation/types/translation.Primary/page.mdx)
- [Query](../../translation/types/translation.Query/page.mdx)
- [RawMessageFormat](../../translation/types/translation.RawMessageFormat/page.mdx)
- [ReadonlyPrimary](../../translation/types/translation.ReadonlyPrimary/page.mdx)
- [Scalar](../../translation/types/translation.Scalar/page.mdx)
- [TranslationStatisticsOutput](../../translation/types/translation.TranslationStatisticsOutput/page.mdx)
